Author’s response to reviews

Title: RENAL TUBERCULOSIS MIMICKING RENAL CELL CARCINOMA: ABOUT A CASE REPORT

Authors:

kays chaker (chakerkays@gmail.com)
Marouene Chakroun (marouenechakroun@gmail.com)
Maroua Gharbi (marouagharb@gmail.com)
Mohamed Chebil (mohamed.chebil.uro@gmail.com)

Version: 1 Date: 27 Mar 2019

Author’s response to reviews:

1- The image in Figure 1 has been optimized for better resolution, contrast and sharpness.

2- Figure 3: The granuloma lesion, including Langhans giant cells with hematoxylin and eosin (×200).

3- The reference number 5 is the same as reference number 1 and one of them needs to be deleted and replace their respective numbers in the text. It’s DONE